\item The Final Exam will take place on Wednesday April 27, 7-10pm in person at EX200.
\item Our TAs Sebastian and Shuyang will hold extra pre-exam office hours, in their usual zoom rooms. Sebastian on Wednesday April 20 and Friday April 22 at 11am-1pm at \href{https://utoronto.zoom.us/j/87218810932}{Sebastian's Zoom} (password vchat), and Shuyang on Monday, Tuesday, and Wednesday April 25-27 at 11am-1pm at \href{https://utoronto.zoom.us/j/83428997680}{Shuyang's Zoom} (password vchat).
\item I will hold my regular office hours on Tuesday April 12, at 9:30-10:30am, I will skip my regular office hours on April 19 and 26, and I will hold extra office hours on Monday, Tuesday, and Wednesday April 25-27 1:30-4pm. All these in person at BA 6178 and online at \url{http://drorbn.net/vchat}.
\item Material: {\em Everything} excluding the material on Maxwell equations, with very light further emphasis on the material not covered in the previous term tests. Very roughly, one third of the questions will come straight from class material, one third from HW, and the last third will be fresh questions.
\item No outside material will be allowed other than stationery, minimal hydration and snacks, and stuffed animals.
\item The format will be ``Solve 7 of 7'', or maybe ``6 of 6'' or ``8 of 8'', in 3 hours.
\item To prepare: Consider last year's final assessment and last year's ``reject'' questions, yet bear in mind that these are all ``fresh'' questions as last year the test was online. Much more important: make sure that you understand every single bit of class material!
